Author's response to reviews

Title: Management of high-risk patients with hypertension and left ventricular hypertrophy in Germany: differences between cardiac specialists in the inpatient and outpatient setting

Authors:

- Heinz Voeller (heinz.voeller@klinikamsee.com)
- Frank Sonntag (f.j.sonntag@t-online.de)
- Joachim Thierry (thiery@medizin.uni-leipzig.de)
- Karl Wegscheider (karl.wegscheider@t-online.de)
- Friedrich C. Luft (luft@fvk-berlin.de)
- Kurt Bestehorn (kurt_bestehorn@msd.de)

Version: 4 Date: 20 September 2006

Author's response to reviews:

The last section of the discussion has been separated as conclusion, as requested.
A new version of figure 3 (without large white margins, but with identical contents), has been added.